Author's response to reviews

Title: Association of Methylentetraydrofolate Reductase (MTHFR) 677 C > T gene polymorphism and homocysteine levels in psoriasis vulgaris patients from Malaysia: a case-control study.

Authors:

Siaw C LIEW (siawcheok_liew@imu.edu.my)
Esha DAS GUPTA (eshadas_gupta@imu.edu.my)
Shew F WONG (shewfung_wong@imu.edu.my)
Nagarajah LEE (nagarajah@oum.edu.my)
Najeeb SAFDAR (najeebkishwar@yahoo.com)
Adawiyah JAMIL (adda_jamil@yahoo.com)

Version: 3 Date: 8 December 2011

Author's response to reviews: see over
Siaw Cheok LIEW,
Department of Postgraduate Studies and Research
International Medical University,
No 126 Jalan Jalil Perkasa 19,
Bukit Jalil, 57000 Kuala Lumpur, Malaysia.
Email : siawcheok_liew@imu.edu.my

Editor
Dr Nehme Gabriel,
Nutrition Journal

Dear Dr Gabriel,

RE: Association of Methylentetraydrofolate Reductase (MTHFR) 677 C > T gene polymorphism and homocysteine levels in psoriasis vulgaris patients from Malaysia: A Case-Control study.

Siaw C LIEW, Esha Das-GUPTA, Shew F WONG, Nagarajah LEE, Najeeb SAFDAR and Adawiyah JAMIL.

We hereby submit our revised manuscript online titled “Association of Methylentetraydrofolate Reductase (MTHFR) 677 C > T gene polymorphism and homocysteine levels in psoriasis vulgaris patients from Malaysia: a case-control study” for your consideration for publication in Nutrition Journal as a research paper.

The revisions were made according to the request by the Independent Editorial Production Team at the proofing stage.

The revisions made are stated as below:

1. The authors’ list in the manuscript were revised according to the authors’ list in the submission system.
2. The abstract uploaded was revised according to the one in the submission system.
3. The tables were revised using the tables tool and orientated in the portrait page layout.
4. The title was revised to follow the sentence case both in the manuscript and the submission system.
5. The keywords were removed as per request.
6. The previously highlighted texts were removed from all the highlighting made.
7. The abbreviations were formatted into a paragraph.
8. The references were cited in full and all “et. al” were removed.
9. The doi numbering was removed and replaced with the full citation.
10. The legends were removed from the image file and were placed at the end of the manuscript. The figures were cropped as closely as possible to minimise the white spaces around the image.
The revised manuscript conforms to the journal style.

We look forward to hearing a favourable reply from you regarding this manuscript.

Thank you.

Yours sincerely,

SC LIEW

Siaw Cheok LIEW
7th December 2011